Author's response to reviews

Title: Acral Post Traumatic Tumoral Calcinosis in Pregnancy: A Case Report

Authors:

Nick Hutt (nickhutt2@hotmail.com)
Paul Baghla (paulbaghla@blueyonder.co.uk)
Vivek Gulati (vivek0@hotmail.com)
Philip Pastides (ppastides@hotmail.com)
Mike Beverly (mikebeverly@hotmail.com)
Wassim Bashir (wassimbashir24@hotmail.com)

Version: 3 Date: 6 July 2010

Author's response to reviews:

The abstract is now broken down under the appropriate subheadings
figure legends have been removed from the main text
the cover letter is removed from the main text
the competing interest and patient consent is now just before the references
the author contribution is now after the reference section